Author's response to reviews

**Title:** The fatty acid binding protein 7 (FABP7) is involved in proliferation and invasion of melanoma cells

**Authors:**

Ana Slipicevic (Ana.Slipicevic@rr-research.no)
Kjersti Jørgensen (Kjersti.Jorgensen@rr-research.no)
Martina Skrede (martina.skrede@gmail.com)
Anne Katrine Ree Rosnes (Anne.Katrine.Rosnes@rr-research.no)
Gunhild Trøen (Gunhild.Troen@rr-research.no)
Ben Davidson (ben.davidson@medisin.uio.no)
Vivi Ann Flørenes (Vivi.Ann.Florenes@radiumhospitalet.no)

**Version:** 6 **Date:** 26 September 2008

**Author's response to reviews:** see over
Dear Editor

Oslo, 26.09.2008

Please find enclosed the final version of the manuscript by Slipicevic et al. entitled ”The fatty acid binding protein 7 (FABP7) is involved in proliferation and invasion of melanoma cells”.

The manuscript has now been formatted after the requirements of The BioMed Central Editorial Production Team

All authors have read and approved the final manuscript.

Yours sincerely,

Ana Slipicevic, MSc
Dept. of Pathology,
The Norwegian Radium Hospital, HF
Montebello,
0310 Oslo,
Norway

Tel 47-22 93 55 88
Fax 47-22 73 01 64
Email ana.slipicevic@rr-research.no